Author's response to reviews

Title: The clinical meaningfulness of ADAS-Cog changes in Alzheimer's disease patients treated with donepezil in an open-label trial

Authors:

Kenneth Rockwood (Kenneth.Rockwood@Dal.ca)
Sherri Fay (Sherri.Fay@cdha.nshealth.ca)
Mary Gorman (mgforman@ns.sympatico.ca)
Daniel Carver (domdj@cdha.nshealth.ca)
Janice E Graham (Jaince.Graham@Dal.ca)

Version: 5 Date: 28 August 2007

Author's response to reviews: see over
Dear Dr. Le:

My co-authors are pleased that our manuscript will be published in BMC Neurology. As per your request, the following formatting changes have been applied to the manuscript:

p.1  The authors are now cited in the manuscript exactly as per their order and appearance in the submission system (Janice Graham change to Janice E. Graham).

p.3  [1-2] changed to [1,2]


p.12  [27-28] changed to [27,28]

p.13  The abbreviations have been separated by commas and semicolons within one large paragraph.

p.13  The statement “He is part owner of www.dementiaguide.ca, a website…” now reads “He is part owner of DementiaGuide Inc., which runs a website…”

p.18  The title ‘Figures’ has been changed to 'Figure legends'

p.20  We accepted your suggestion and changed the legend for Table 2 to read as follows:
*Higher scores indicate better performance (the expected direction is positive), otherwise lower scores indicate improvement.

I look forward to receiving the galley proofs.

Yours sincerely,

Kenneth Rockwood
Professor of Medicine (Geriatric Medicine & Neurology) and
Kathryn Allen Weldon Professor of Alzheimer Research
Geriatric Medicine Research Unit
Dalhousie University
Suite 1421, Camp Hill Veterans’ Memorial Building
5955 Veterans’ Memorial Lane
Halifax, NS B3H 2E1 Canada